Author's response to reviews

**Title:** Soluble egg antigen of Schistosoma Haematobium induces HCV replication in PBMC from patients with chronic HCV infection

**Authors:**

Mostafa K El Awady (mkawady@yahoo.com)
Samar S Youssef (youssefsamar@hotmail.com)
Moataza H Omran (moataza13@yahoo.com)
Ashraf A Tabll (ashraftabll@yahoo.com)
Wael T El Garf (elgart@yahoo.com)
Ahmed M Salem (ahmedsalem47@hotmail.com)

**Version:** 9  **Date:** 28 April 2006

Author's response to reviews:

1. Title & Abstract are identical in the manuscript and in the submission system. The abstract in the submission system has been sub headed Background, Results and Conclusion.
2. Title: full stop is removed.
3. Author list: initial is corrected to "K".
4. References - References has been cited in the text using consecutive normal case numbers in square brackets instead of superscript numbers.